Author's response to reviews

Title: The fetuses-at-risk approach: clarification of semantic and conceptual misapprehension

Authors:

K S Joseph (ksjoseph@dal.ca)

Version: 2 Date: 7 March 2008

Author's response to reviews:

Dear Editor,

The Reviewer did not require any changes to be made to the submitted manuscript. The Editorial request for providing a 'conflict of interest' statement has been addressed in the revised manuscript. The manuscript conforms with the Journal format for 'Correspondence' type articles. Minor edits to the text have also been made.

Yours sincerely,

K.S. Joseph MD, PhD
Dalhousie University and the IWK Health Centre